Author’s response to reviews

Title: Metformin associated lactic acidosis: a case series of 28 patients treated with sustained low efficiency dialysis (SLED) and long-term follow-up

Authors:
Andrea Angioi (andrea.angioi@gmail.com)
Gianfranca Cabiddu (gianfrancacabiddu@aob.it)
Maura Conti (mauraconti@aob.it)
Gianfranco Pili (gianfrancopili@aob.it)
Alice Atzeni (aliceatzeni@aob.it)
Valeria Matta (valeriamatta@aob.it)
Riccardo Cao (riccardocao@aob.it)
Matteo Floris (matteofloris@aob.it)
Marco Songini (marcosongini@aob.it)
Maria Franca Mulas (mariafrancamulas@aob.it)
Mitchell Rosner (MHR9R@hscmail.mcc.virginia.edu)
Antonello Pani (antonellobali@oab.it)

Version: 3 Date: 26 Feb 2018

Author’s response to reviews:

1. "We would also like to ask for you to provide more justification for the contributions of GC as currently they are not listed within the author contributions”.

- We modified the "Authors' contributions” as follows: "M.C., G.P., A.A.2, V.M., P.M., R.C., G.C. reviewed and edited the manuscript". 
2. Please edit the statement ‘asked for a written consent to every patient (or its relatives)’ to ‘asked for a written consent to every patient (or their relatives)’

- we modified the manuscript accordingly

3. Please confirm why the relatives gave consent for some patients?

- we added the following sentence: "at the admission, we asked for a written consent to every patient (or their relatives when patients were not able to give a consent)".

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

- we submitted the manuscript in accordance with the Journal guidelines